Author's response to reviews

Title: Interns' knowledge of clinical pharmacology and therapeutics after undergraduate and on-going internship training in Nigeria

Authors:

Kazeem A Oshikoya (med_modhospital@yahoo.com)
Idowu O Senbanjo (senbanjo001@yahoo.com)
Olufemi O Amole (femiamole@yahoo.co.uk)

Version: 3 Date: 14 July 2009

Author's response to reviews: see over
Response to reviewers’ comments

Reviewer: Simon Maxwell.

Many thanks again for your recent comments on our manuscript. We have modified the title of the manuscript to read “Interns’ knowledge of clinical pharmacology and therapeutics after undergraduate and on-going internship training in Nigeria: a pilot study”. The study has, to an extent, assessed the interns’ knowledge of CPT self-rated confidence. However, we did not assess their prescribing skills which we have acknowledged as one of the limitations of the study. The authors intend to conduct a national study in the nearest future to address all our limitations and those currently raised by the reviewer.